

Citation: Duong D, Johny AR, Ledgister Hanchard S, Fortney C, Flaharty K, Hellmann F, et al. (2024) Comparison of clinical geneticist and computer visual attention in assessing genetic conditions. PLoS Genet 20(2): e1011168. https://doi.org/ 10.1371/journal.pgen.1011168

Editor: Gregory S. Barsh, HudsonAlpha Institute for Biotechnology, UNITED STATES

Received: September 8, 2023

Accepted: February 5, 2024

Published: February 27, 2024

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: Data are available in the main paper and supplementary files; additional data and code are available at: https://github.com/ datduong/tobii-eye-track-syndromic-faces (to compare heatmaps) and https://github.com/ datduong/classify-syndromic-faces (for classifier and occlusion maps).

Funding: This research was supported in part by the Intramural Research Program of the National Human Genome Research Institute, National Institutes of Health. This funding supported **RESEARCH ARTICLE**

Comparison of clinical geneticist and computer visual attention in assessing genetic conditions

Dat Duong¹[©], Anna Rose Johny²[©], Suzanna Ledgister Hanchard¹[©], Christopher Fortney³[©], Kendall Flaharty¹, Fabio Hellmann⁴, Ping Hu¹, Behnam Javanmardi⁶², Shahida Moosa^{5,6}, Tanviben Patel⁶¹, Susan Persky³, Ömer Sümer⁶⁴, Cedrik Tekendo-Ngongang¹, Hellen Lesmann², Tzung-Chien Hsieh², Rebekah L. Waikel¹, Elisabeth André⁶⁴, Peter Krawitz², Benjamin D. Solomon¹*

1 Medical Genetics Branch, National Human Genome Research Institute, Bethesda, Maryland, United States of America, 2 Institute for Genomic Statistics and Bioinformatics, University Hospital Bonn, Rheinische Friedrich-Wilhelms-Universität Bonn, Bonn, Germany, 3 Social and Behavioral Research Branch, National Human Genome Research Institute, Bethesda, Maryland, United States of America, 4 Chair for Human-Centered Artificial Intelligence, University of Augsburg, Augsburg, Germany, 5 Division of Molecular Biology and Human Genetics, Stellenbosch University, Stellenbosch, South Africa, 6 Department of Medical Genetics, Tygerberg Hospital, Tygerberg, South Africa

These authors contributed equally to this work.
* solomonb@mail.nih.gov

solomonb@mail.nin.gov

Abstract

Artificial intelligence (AI) for facial diagnostics is increasingly used in the genetics clinic to evaluate patients with potential genetic conditions. Current approaches focus on one type of AI called Deep Learning (DL). While DL- based facial diagnostic platforms have a high accuracy rate for many conditions, less is understood about how this technology assesses and classifies (categorizes) images, and how this compares to humans. To compare human and computer attention, we performed eye-tracking analyses of geneticist clinicians (n = 22) and non-clinicians (n = 22) who viewed images of people with 10 different genetic conditions, as well as images of unaffected individuals. We calculated the Intersection-over-Union (IoU) and Kullback-Leibler divergence (KL) to compare the visual attentions of the two participant groups, and then the clinician group against the saliency maps of our deep learning classifier. We found that human visual attention differs greatly from DL model's saliency results. Averaging over all the test images, IoU and KL metric for the successful (accurate) clinician visual attentions versus the saliency maps were 0.15 and 11.15, respectively. Individuals also tend to have a specific pattern of image inspection, and clinicians demonstrate different visual attention patterns than non-clinicians (IoU and KL of clinicians versus non-clinicians were 0.47 and 2.73, respectively). This study shows that humans (at different levels of expertise) and a computer vision model examine images differently. Understanding these differences can improve the design and use of AI tools, and lead to more meaningful interactions between clinicians and AI technologies.

authors who are NHGRI staff, and includes salary support and research expenses (these authors: DD, SLH, CF, KF, PH, TP, SP, CT-N, RLW, BDS). There is no specific grant number for this intramural funding. See: https://www.genome.gov/ about-nhgri/Division-of-Intramural-Research. The funder did not play any role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: BDS is the Co-Editor-in-Chief of the American Journal of Medical Genetics, and receives textbook royalties from Wiley.

Author summary

Artificial intelligence (AI) is increasingly used in medicine. In clinical practice, medical geneticists often use AI tools to help them examine the facial features of a patient who might have a genetic condition. While these tools are very popular, less is understood about how they work, and which parts of the face are most important to the AI tools, especially compared to medical geneticists. To address this, we performed a study where medical geneticists (as well as non-clinicians) looked at pictures of people with and without genetic conditions. We used eye-tracking tools to visualize which parts of the images the medical geneticists preferentially looked at. We compared this to the parts of the image that were the most important to the AI tools. We found that the medical geneticists and the AI tools tended to pay attention to very different features. We also found that medical geneticists and non-clinicians look at different parts of the images. Understanding how AI tools examine images compared to clinicians can help ensure the tools function properly, and could also help perform tasks like alerting medical geneticists to important clinical findings that could otherwise be overlooked.

Introduction

Deep learning (DL), one subtype of artificial intelligence (AI), is extensively and increasingly employed in many biomedical areas. For example, generative language models show promise for analysis of medical records and related data [1]. For genomic data, DL methods been used for protein modeling and variant classification [2,3]. In medical genetics, clinicians and researchers often employ DL-based approaches to generate differential diagnoses based on facial images of individuals with possible genetic conditions [4,5].

Despite controversies and unsettled questions, healthcare is poised for major near-term changes as AI is implemented into workflows [1]. One crucial issue involves understanding how AI tools perform and how "choices" made by AI compare with those made by humans, including people with different levels of expertise [6–8]. For example, generative language models may sometimes present coherent but incorrect or inconsistent information [9]. As another example, an analysis of DL to assess chest X-rays for signs of COVID-19 showed that classifiers can be highly accurate compared to humans but may rely heavily on confounding information, such as radiographic clues outside of the lung fields [10]. Interestingly, though DL analysis of facial images has been a major application in medical genetics, relatively little work has been done to compare how clinicians view images compared to DL models [11].

To further explore this topic, we used an eye-tracking device to measure human visual attention when inspecting images of individuals with and without genetic conditions. Clinicians in our research group first selected images of several relatively common and recognizable genetic conditions, which were then viewed by other genetic medical experts as well as non-clinicians. We further trained a DL classifier to predict the ground-truth labels of these images and computed the corresponding saliency map for each image. Saliency maps allow the visualization of which parts of an image are important to the model when "deciding" how to classify an image.

First, we examined how well the visual attention maps of genetic medical experts (termed "clinicians" here) align with the classifier saliency maps. Second, we compared the visual activity of clinicians and non-clinicians. The comparison metrics for the visual maps are Intersection-over-Union (IoU) and Kullback–Leibler divergence (KL). IoU binarizes the visual maps and measures the amount of overlap between two heat maps, whereas KL treats the heat maps as continuous probability distributions and calculates the divergence between these two distributions. For both metrics, we found that saliency maps differ substantially from regions of human attention and that clinicians and non-clinicians assess images differently.

Results

Data collection

In this paper, we selected 10 images for the eye-tracking experiments, one for each of the following conditions: 22q11.2 deletion syndrome (22q11DS, OMIM #188400), Beckwith-Wiedemann syndrome (BWS, OMIM #130650), Cornelia de Lange syndrome (CdLS, OMIM #122470), Down Syndrome (DS, OMIM #190685), Kabuki syndrome (KS, OMIM #147920), Noonan syndrome (NS, OMIM #163950), Prader-Willi syndrome (PWS, OMIM #176270), Rubinstein-Taybi syndrome (RSTS1, OMIM #180849), Wolf-Hirschhorn syndrome (WHS, OMIM #194190), and Williams syndrome (WS, OMIM #194050). Next, six additional images of unaffected individuals selected to represent images of similarly aged individuals with similar photographic poses were also chosen as the control group (see more details in Materials and Methods and references for image sources in <u>S1 Text</u>). In summary, there are 16 images to be tested in our eye-tracking experiment.

Human participant performance

We divided the human participants into two main groups: 22 clinicians (15 from NIH and seven from Bonn) and 22 non-clinicians. Table 1 compares the accuracy of the two groups at

Table 1. Summary of results for each image viewed. For a specific image, we count the number of clinicians and non-clinicians who correctly and incorrectly identified that this image is of an affected or unaffected individual. Column *Identified condition name* indicates the number of clinicians and non-clinicians who correctly identified the specific condition. As eye-tracking results for seven individual images had to be discarded for technical reasons, not every image has a complete data from 22 clinicians and 22 non-clinicians. The images are listed in the order shown to the participants.

	Clinicians				Non-clinicians		
	Correct	Incorrect	Identified condition name	Correct	Incorrect	Identified condition name	
ws	17	5	4	13	9	3	
Unaffected	22			19	3		
RSTS1	21		3	21	1		
Unaffected	19	3		14	8		
WHS	20	2		19	3		
Unaffected	22			20	2		
CdLS	22		10	22		5	
DS	22		18	21	1	1	
Unaffected	15	6		15	6		
KS	19	3	11	13	9	2	
NS	21		9	21			
Unaffected	18	4		18	4		
22q11DS	21	1	3	19	3		
PWS	16	5		10	12		
Unaffected	14	8		19	3		
BWS	8	13	1	5	17		

abbreviations: BWS: Beckwith-Wiedemann syndrome; CdLS: Cornelia de Lange syndrome; DS: Down Syndrome; KS: Kabuki syndrome; NS: Noonan syndrome; PWS: Prader-Willi syndrome; RSTS1: Rubinstein-Taybi syndrome; WHS: Wolf-Hirschhorn syndrome; WS: Williams syndrome

https://doi.org/10.1371/journal.pgen.1011168.t001



Fig 1. Example image illustrating the preprocessing step of visual heat maps for subsequent analyses. (a) Original image of an individual with Down syndrome (DS). (b) For this DS image, conditioned on the group of successful clinicians, we average the default attention heat maps from the eye-tracking experiments. (c) We removed the common visual signals (See <u>\$27 Fig</u>). (d) Finally, we smooth the image with cv2.boxFilter and increase the color intensity. Here, compared to (b), in (d) we can better observe the high visual interest for a larger orbital region (not just the center of the eyes), as well as attention to the ears, which, though barely visible in this image, can be notable features on physical examination for this condition. The image is freely available for reuse without restriction, courtesy: National Human Genome Research Institute (www.genome.gov).

identifying whether an image shows a person affected or unaffected with a genetic condition. Overall, the clinician group was more accurate than the non-clinician group at recognizing whether an image showed an affected person (correct identification was 85.6% for clinicians versus 76.9% for non-clinicians, p = 0.0032 by Chi-square). In further responding to a question about what specific genetic condition the image might depict, non-clinicians were, unsurprisingly, usually unable to identify the disease names. However, some non-clinicians are familiar with specific conditions based on their work, such as involvement in genetic research.

For the rest of the analysis, we divide all the participants into four subgroups: clinicians and non-clinicians who correctly or incorrectly identified that an image represents a person with a genetic condition. We use the terms *successful* and *underperforming clinicians* and *non-clinicians* to refer to these subgroups, respectively. For two different test images, the same participant may fail to recognize the disorders in just one of these images. Hence, for two different images, the groups of successful clinicians (and likewise non-clinicians) may not have the same participants.

To understand the average visual behavior of a specific participant group (e.g., successful clinician group), we computed the average heat map for each test image over all the participants within the group (see Fig 1 and the Materials and Methods section for further details about our data preprocessing approach). Two noise-threshold levels were applied, where a pixel value below the threshold is set to 0. The lower noise-threshold removes spurious visual signals. The higher noise-threshold removes a large proportion of the signal; thus, we would analyze only facial regions with the highest visual attention (Figs 2 and S1–S10).

Deep learning classifier performance

Since datasets and/or source codes of widely clinically used DL approaches in medical genetics cannot be easily shared (e.g., for the tool available via Face2Gene, https://www.face2gene.com/), we trained our own image classifier in this paper. Our model was trained to predict (identify) 11 different labels (10 diseases and one unaffected group). When considering unaffected and affected as the only two label choices, our classifier achieves 100% accuracy. Related to the model's identification of specific conditions, our classifier correctly labeled 13 out of the 16 test images with the specific genetic condition that person has; averaging over these 13 images, the prediction probability of the ground truth label is 0.7853. The three misclassified affected images are CdLS, PWS, and RSTS1. For these three images, the predicted probabilities of the



Fig 2. Application of noise-thresholds to human heat maps and coverage-thresholds to model saliency maps. Eye gaze heat maps are shown for the 22q11DS image used in our experiment. The left-most column shows the original average heat maps for the successful clinicians and non-clinicians. Low noise-threshold removes possibly spurious visual interests, and then the higher noise-threshold better indicates facial regions with the most prominent visual attentions. Two coverage-thresholds were applied to the model saliency map of this image; the objective is to match the number of positive pixels in the saliency maps with respect to the clinician visual attentions at the low and high-noise thresholds, respectively. Pixels below the coverage-threshold were set to zero, and the remaining number of positive pixels are approximately equal to that of the corresponding human visual map. The image is freely available for reuse without restriction, courtesy: National Human Genome Research Institute (www.genome.gov).

ground truth label and incorrect label were: CdLS (0.3196 vs 0.3285 misclassified as NS), PWS (0.3383 vs 0.4690 misclassified as WS), and RSTS1 (0.0005 vs 0.7976 misclassified as NS). Because the classification accuracy of RSTS1 is low, we excluded this image in the saliency map analysis in the next section (S11 Fig contains the analysis for all the affected images). See more details in the Materials and Methods section below and S1 Text).

Evaluating human visual attention versus saliency map

For a specific input image, one can extract the corresponding saliency map from an image classifier (see details in the Materials and Methods section, including under the *Classifier and saliency maps* heading). This saliency map indicates the regions of the image that the model deems to be relevant for the prediction of the ground truth label. Our goal is to observe (1) which regions of an image affect the prediction probability of the ground-truth label and (2) how well these regions align with the areas most viewed by human clinicians. Hence, the saliency maps were made with respect to the ground-truth labels, whether or not the model correctly identifies these ground-truths.

For the analyses, we treat the saliency maps as if they were heat maps. We note the following key observation. Human visual heat maps produced via the Tobii eye-tracking software are partially binarized; that is, most of the visual map is blank (i.e., zero pixel value), with only certain key areas showing a color gradient (i.e., positive pixel value) to indicate key visual activity. We applied the occlusion saliency method to compute the positive contribution of every patch of pixels of an image toward the prediction of its ground-truth label. Hence, an occlusion saliency map typically contains a larger number of positive pixels than the human visual heat map. Highly unequal numbers of positive pixels between two inputs can lead to unreliable IoU and/or KL metrics. Hence, we applied a coverage-threshold to the saliency map, where pixel value below the coverage-threshold is set to 0 and the remaining number of nonzero pixels is equal to that of the human visual map (Fig 2).





For a specific test image of an affected individual, we computed the IoU score and KL divergence between the saliency map and the average heat map of the successful clinician group at both low and high noise-thresholds. At each noise-threshold of the human visual attention, we used grid-search to find the corresponding coverage-threshold for the saliency map, so that the number of nonzero pixels in the saliency map is approximately equal to that of the human visual attention. To compute IoU, pixel values above zero were converted into 1; and to compute KL, the pixel values were rescaled into a proper probability distribution. More details are explained in the Materials and Methods, with results shown in Fig 3 (see also S12 Fig).

Fig 3 shows the IoU and KL metrics averaged over the 9 affected images (excluding RSTS1 for reason mentioned above). We observed significant differences between human visual attention and what a DL classifier considers important. For example, the average IoU metric between successful clinicians versus the saliency map were 0.15 and 0.10 at low and high noise-threshold, respectively. This IoU metric is much less than the metric comparing our clinicians and non-clinician groups (see below). In other words, human heat maps show that our participants tended to look at very different regions than the classifier per the models we examined. The KL metric also indicates the same trend, where we observe a high KL value, indicating a large difference between human visual interest and model saliency maps.

Comparing groups of human participants

We also compared the successful clinicians and non-clinicians for each of the 16 test images, and then summarized the visual attention differences over all these images. Here, we do not apply a coverage-threshold because heat maps of two participant groups may naturally have very different coverages. For example, it is entirely possible that, compared to the experts, non-clinicians' visual activities are more dispersed since they may not recognize the signs of the genetic conditions. In **Figs 2 and S1–S10**, qualitatively speaking, the visual attentions of these clinicians and non-clinicians are more similar at a lower noise-threshold, where only spurious visual signals were removed, and later become more different at a higher noise-threshold, where only regions with high visual interests were retained. (Note that figures in **S1–S10** and **S13–S23 Figs** are shown as segmented versions of the original images are provided in the **S1 Text** file. Segmentation for the visualization was performed via previously published methods. [12]). Thus, as shown in **Figs 2 and S1–S10**, although these clinicians and non-clinicians correctly recognize affected versus unaffected individuals, visually speaking, on average these two participant groups do not appear to display the same trend when assessing the images.



Fig 4. Comparison of visual heat maps of successful (correct) human participants at different noise-threshold levels. IoU and KL metric compare the visual heat maps of successful clinicians versus successful non-clinicians. When removing only low spurious signal ("threshold low"), the two groups display more similar visual interests. However, when considering only regions with high visual interests ("threshold high"), the two groups differ more drastically (average IoU drops from 0.47 to 0.34, and average KL increases from 2.73 to 4.93). The images are listed in the order in which images were shown to the participants.

Fig 4 numerically compares the visual interests of successful clinicians versus successful non-clinicians. Here, the result aligns with our qualitative inspection in **Fig 2**. When excluding low spurious signals with a low noise-threshold, these two subgroups show similar eye-gaze interests (IOU = 0.47, and KL = 2.73). However, these two subgroups become more different when considering only the highest visual attention (IOU dropped to 0.34, and KL increased to 4.93). A similar trend is observed when we analyzed on just the set of affected and unaffected images (**S12 Fig**).

We further compare the successful and underperforming clinician group. Fig 5 shows that, on average, these two subgroups of clinicians do not have similar visual interests. The same trend is seen when stratifying the non-clinicians according to their accuracy. Conditioned only on participants who misclassified the images, Fig 5 also shows visual attention differences between clinicians and non-clinicians. Hence, when participants misclassify images, they have distinct ways of inspecting the images that do not appear to be similarly influenced by some common confounders in the images (e.g., hairstyles, facial expressions, or earrings that could affect a participant's decision). Overall, there seems to be more similar visual attention when a participant (whether a clinician or not) can correctly identify that a person is affected by a genetic condition; for example, the IoU values in Fig 4 are higher than those in Fig 5.

We also examined aggregated heat maps produced by each of the clinicians and non-clinicians and evaluated whether there were different visual patterns when the same participant examined affected or unaffected images. As shown in **Figs** <u>\$24-\$26</u> show that participants had different patterns (ways) of assessing images.

Finally, we examined the AOIs that corresponded to specific dysmorphic features in the images of affected individuals (S14–S23 Figs). As each test image has more than one AOI that may be important to the underlying conditions, metrics like duration-of-fixation and time-to-first-whole-fixation for any single AOI can have high standard deviations, making it difficult to observe statistical significance. Overall, we did not detect a specific pattern for these AOIs that differentiated the categories of participants.

Discussion

In clinically oriented analyses, the performance of DL and other AI models have traditionally been compared to that of human experts, such as evaluating how well DL models versus humans recognize the presence of metastatic cancer from lymph node sections or assess chest X-rays from inpatient and urgent care settings [13,14]. There has overall been less attention to



Fig 5. Additional comparisons of human participants. Images are sorted based on the IoU (top row) and KL (bottom row), which compare the heat maps between different groups of participants. A low noise-threshold was applied to remove spurious visual activity from the average heat map of each group. These three comparisons show lower similarity scores than that of successful clinicians versus successful non-clinicians (IoU = 0.47, and KL = 2.73 in Fig 4). Clinicians tend to show more similar visual interests for WS, PWS, BWS, and unaffected images (left column). This trend is not seen for non-clinicians (middle column).

comparing the *processes* of computers and humans when performing tasks. One reason for this involves the fact that, when DL classifiers emerged, they were not originally designed, nor were there available saliency approaches, to enable easy visual explanation of their processes. More recently, saliency methods were developed independently of the DL classifiers, and have been explored in a variety of biomedical contexts, including related to multiple types of radiologic studies (e.g., computed tomography, magnetic resonance imaging, retinal imaging, and X-rays), electrocardiograms, and in a variety of pathology analyses [15]. These studies primarily focused on comparing saliency results to the portions of biomedical images that had been annotated to contain the most clinically relevant data, such as specific cell types in pathology images, but paid less attention to the overall process [16]. This overall process may be important to consider to investigate questions like pertinent positives and negatives, as well as to search for potential confounders.

Against this background, our study has two key results. First, human and computer attention differs substantially when evaluating images of individuals with potential genetic conditions. This does not mean that one is superior to the other, but these types of analyses and metrics may be helpful in future studies. For example, methods to compare human and computer attention can be used to explore potential confounders in AI-based analyses, or to develop methods that improve the accuracy or applicability of AI tools. As AI is increasingly adopted in clinical scenarios, such studies will be critical to assess model performance. For generalizability, such future studies would need to be much larger, both in terms of the number of participants and the number of images and genetic conditions included. Second, clinicians and non-clinicians exhibit different gaze behaviors when assessing images. This is not surprising, but quantifying these behaviors using methods like these may be helpful for activities such as ascertaining which phenotypic characteristics may be diagnostically important but which are frequently overlooked. Again, as AI support enters more and more clinical areas, these types of analyses may point to specific ways to augment the relationship between clinicians and AI tools. For example, data from extremely high-performing clinicians in human/classifier comparison experiments may be useful to design the education of less experienced clinicians or trainees, as well as AI tools [17].

This study has several limitations. These include the number of participants and images viewed. The images also represent heterogeneous genetic conditions, and eye-tracking behavior may be affected by certain aspects of a particular image. While we grouped clinicians and non-clinicians into separate categories, there is varying experience and expertise within these groups, and differences in behavior between individuals. Additionally, our analyses focused on metrics like human visual attention, which may not reflect what is most important to a person evaluating an image. For example, an expert clinician may immediately perceive a key visual clue to a diagnosis and may then move on to spend more time observing less obvious features or searching the image for subtle clues.

From the DL perspective, we do not claim that our model is highly accurate or represents the best possible option-our aim was to compare what a DL model weights when analyzing images to what a human looks at, and we intentionally chose images with a range of performance on our DL classifier, as well as images that we subjectively felt would have a range of difficulty for human participants. Along these lines, saliency map analyses were not performed with respect to the correct and incorrect predicted label, separately. Rather, the DL saliency map analyses were performed with respect to the ground-truth labels, whether or not the ground-truths are the model's correct prediction. The only requirement was that our model's predicted probabilities for these ground-truths needed to be larger than zero. Otherwise, the saliency map may not be reliable when the model fails to make any prediction for this ground-truth label [15]. In this current study, the chosen images each have a set of key condition-specific features unique to that specific condition, and each condition is relatively distinct from each other. In future studies, we plan to select a larger number of conditions that have more shared syndromic features among several different conditions (e.g., many different conditions that all involve facial findings like hypertelorism or upslanted palpebral fissures), which would let us better understand and compare the behavior of the model and humans, such as when they both incorrectly classify an image. Overall, while our results show interesting trends, due to small number of test images, caution should be taken that they will generalize to other or all genetic conditions, to larger groups of individuals, or to other DL models or AI-based approaches.

As mentioned above, the results here point to multiple future possibilities, including involving larger numbers of participants and datasets to analyze further how well the results can be extrapolated. Related eye-tracking experiments could be used to explore multiple questions germane to genetics, such as with different data types (e.g., radiological studies or other physical examination features) encountered in clinical practice. Additional work could be done specifically quantifying different classifiers and saliency methods.

Materials and methods

Ethics statement

This study was approved as IRB exempt by the IRBs of the National Institutes of Health (NIH) (Bethesda, Maryland, United States) and the University of Bonn (Bonn, Germany). NIH study ID: 000846; Bonn study ID: 210/23-EP. All participants were adults (the images viewed were of

previously published pediatric individuals); formal consent was not required, with subjects agreeing to participate by taking part in the study.

Data collection

Similar to our previous methods [18,19], we selected publicly available images of individuals between 2 and 18 years of age affected by one of 10 genetic conditions (for reference, we provide the first OMIM #ID for conditions with multiple OMIM # IDs): 22q11.2 deletion syndrome (22q11DS, OMIM #188400), Beckwith-Wiedemann syndrome (BWS, OMIM #130650), Cornelia de Lange syndrome (CdLS, OMIM #122470), Down Syndrome (DS, OMIM #190685), Kabuki syndrome (KS, OMIM #147920), Noonan syndrome (NS, OMIM #163950), Prader-Willi syndrome (PWS, OMIM #176270), Rubinstein-Taybi syndrome (RSTS1, OMIM #180849), Wolf-Hirschhorn syndrome (WHS, OMIM #194190), and Williams syndrome (WS, OMIM #194050), as well as images of unaffected individuals similar in age and ancestral diversity to the affected individuals. The genetic conditions were chosen as they represent relatively common genetic conditions that involve recognizable craniofacial features, and with which geneticists would be expected to be at least somewhat familiar [20]. Two geneticists (BDS, CTN) and a genetic counselor (RLW) on the study team selected the images to represent typical images representing the chosen conditions; images of some conditions were felt to be more difficult to recognize than others. Sixteen images (one image for each of the above conditions, and six images of unaffected individuals) were selected for the eye-tracking experiments. The selected images were at least 720-pixel resolution; they were each standardly cropped, centered, and aligned (e.g., the eyes, nose, and mouth were roughly repositioned at the same coordinates for all images See S1 Text for image sources; due to journal image sharing policies, images in S1-S10 and S13-S23 Figs are shown as segmented versions of the images used in the eye-tracking experiments [12].

Eye tracking experiments

The formatted images were embedded in a screen-based eye-tracking system (Tobii Pro X3-120, Tobii Lab Pro version 1.194.41215; https://www.tobii.com/, Stockholm Sweden). This system uses infrared light to detect pupillary movement, which allows the analysis of gaze activity, such as how long they look at different parts of each image. Eye-tracking experiments took place in two locations: the National Institutes of Health (NIH) (Bethesda, Maryland, United States) and the University of Bonn (Bonn, Germany).

After calibration, each participant viewed the 16 images for seven seconds per image, and answered questions about each image, including whether the image showed a person affected by a genetic condition, and, if so, what condition the person had. After extensive initial testing, we chose seven seconds for the viewing time, as subjective feedback and preliminary assessments showed that this amount of time was sufficient to assess an image but minimized visually revisiting areas of the image in a way that might not further inform the assessment. To minimize head movement or distractions, questions were asked verbally during the eye-tracking portions of the experiment. Responses were documented manually by a study team member.

The NIH cohort included 17 individuals, including physician geneticists (n = 14) and physicians in genetics subspecialty training (n = 3). The Bonn cohort included 29 total individuals, including physician geneticists (n = 2) and physicians in genetics subspecialty training (n = 5), as well as 22 non-clinicians. Some of the non-clinicians in our cohort have some experience through their work in recognizing genetic conditions. For example, some are graduate

students who study applications of AI in genetics, and thus are familiar with genetic conditions, but are not trained clinicians.

Data extraction and analysis

We extracted eye-tracking data for analysis in two main ways. First, we extracted individual heat maps for each participant and image (these heat maps show the relative amount of time each participant looked at different parts of each image). We used the default Tobii software settings, but with two changes to enable our analyses: a) we changed the output eye-tracking radius to 25 pixels, as our preliminary analyses showed that the default 50 pixel radius was insufficiently precise for comparisons; b) we used a more homogeneous color scheme for the heat maps (with the following hexadecimal color codes: high: #943126, medium: #B03A2E, low: #CB4335), which enabled our quantitative analyses to take into account all captured heat map data, including to compare eye-tracking data with classifier saliency maps.

Second, to enable additional analyses that took into account gaze trajectory behavior to analyze the timing of participant gaze [21], a dysmorphologist (BDS) manually drew areas-ofinterest (AOIs) for each image (see S14–S23 Figs). The AOI set included only those features that were listed as having dysmorphic manifestations based on the clinical synopses in OMIM (https://www.omim.org/) and which were also present in the images. For example, if a condition were listed in the clinical synopsis section of OMIM as having a dysmorphic manifestation affecting the eyebrows and that manifestation was present in the image of a person with that condition used in the survey, an AOI was drawn around the eyebrows. Using the Tobii software, we extracted tabular data for analysis based on these defined AOIs.

Prior to analysis, we manually reviewed results and excluded heat maps where eye-tracking data was not recorded (this occurred for all results for one NIH participant, which may have been due to ophthalmologic issues like severe myopia, and seven total other isolated eye-tracking results for unclear reasons).

We observed that standard heat maps primarily showed that the areas around the eyes, nose, and mouth are the regions with the highest attention. These common visual attentions (S27 Fig), which likely reflect standard human behavior, make it difficult to quantitatively compare the visual activity differences between groups unless accounted for. To mitigate this issue, we computed the average heat map for all the clinicians and non-clinicians (separately) over all of the images. We then subtracted this common average gaze pattern (separately for clinicians and non-clinicians) from each heat map. This does not cause us to ignore these areas of common facial attention in our analyses but helps account for typical human behavior when viewing faces. Fig 1 explains our data preprocessing approach.

In our analyses of the images, we focused on four subgroups: clinicians and non-clinicians who correctly or incorrectly identified that an image represents a person with a genetic condition. One interest is whether the untrained intuition of a non-clinician aligns with clinician behavior. In the later sections of our analyses, we use the terms *successful* and *underperforming clinicians* and *non-clinicians* to refer to the clinicians and non-clinicians who correctly and incorrectly recognize the presence of a genetic disease for a given test image, respectively. For two different test images, the same participant may fail to recognize the disorders in both images. Hence, for two different images, the groups of successful clinicians (and likewise non-clinicians) may not have the same participants.

To understand the general behavior of a specific participant group (e.g., successful clinician group), after accounting for the areas of common visual attention, we computed the average heat map for each test image over all the participants. Two different thresholds were applied to remove noise from the heat maps, where pixel values below this threshold are set to 0. The

first, lower noise-threshold is meant to remove spurious visual activities. The second, higher threshold removes a large proportion of the signal; here, we would analyze only facial regions with the highest visual attention. At each noise-threshold, to compare the average heat maps (of two participant groups or against the model), we applied two different metrics: Intersection-over-Union (IoU), and Kullback–Leibler divergence (KL).

Previous work applied IoU to benchmark human annotations against DL model saliency maps for chest X-ray images [15]. IoU is symmetric, ranges between 0 and 1, and is intuitive; that is, when similarity is high, then there is more overlap between two heat maps. We note that IoU requires the pixel values to be binarized (e.g., 0 or 1). Here, at each noise-threshold, nonzero pixels (i.e., pixels higher than the noise-threshold) are set as 1 when computing IoU. This binarization may result in the loss of intensity within both human visual heat maps and model saliency maps. Thus, we applied a second metric. Our second metric (KL) does not binarize the pixel values. Instead, KL treats the heat maps as two-dimensional probability distributions and calculates the divergence between these two distributions [22]. Unlike IoU, KL is not symmetric and ranges from 0 to infinity. We opted for the symmetric implementation of KL; that is, we take the average KL (x,y) and KL (y,x) for two inputs x and y. Suppose we want to compare the average visual attention of a clinician group against a model saliency map, respectively. A KL value far from 0 implies that human visual map and model saliency map are very different.

Standard deviation for the observed IoU (and likewise for KL divergence) was computed via bootstrapping. Finally, we applied random effects meta-analysis (we refer to this as "RE Model" in **Figs 3, 4, 5, S11, S12, and S26**) to summarize the behavioral differences over many images between the two groups of participants (e.g., successful clinicians versus successful non-clinicians).

Classifier and saliency maps

Separate from the training and validation set, our test set includes the 16 images used in eyetracking experiments and 2–4 additional images for several selected conditions. With the remaining images in the dataset (not containing any test images), we applied 5-fold cross validation to train 5 different models. For each iteration of the 5-fold cross validation, we finetuned EfficientNet-B4 pre-trained on ImageNet with respect to the other 4 folds and validate it on the left-out fold. EfficientNet-B4 was chosen because relatively it is one of the models requiring a relatively low number of parameters while still obtaining high accuracy on ImageNet [23]. Each image was resized to 448 x 448 pixels, and cross-entropy loss with equal weights for the diseases was used. Grid-search was performed to find a single set of hyperparameters that works best on average over all the 5-folds. We found the following hyperparameters to work best: batch size 64, dropout rate 0.2, and learning rate 0.00003 for Adam optimizer [24]. Training scripts are provided in the GitHub links (see Data and code availability section). **Table A** in <u>S1 Text</u> shows the top-k accuracy (acc@k) on the validation set.

After training 5 models from 5-fold cross validation, we built an ensemble classifier by averaging the predictions of these models for a test image. **Table B** in **S1 Text** shows the model performance for the images used in eye-tracking experiments (**Table C** in **S1 Text** shows performance on additional selected test images). Again, we aimed to compare which portions of an image were important to a DL classifier versus human eye-tracking attention, and different models would be expected to return different results; limitations for our classifier are that [1] our model misclassifies some images and [2] even when the predictions are correct, the predicted probabilities for the ground-truth labels may not be close to 1. The occlusion saliency method with box size 20 x 20 pixels and stride 10 x 10 was used to produce the saliency maps [25]. The intuition is to remove a 20 x 20 pixel box from the image (e.g., set the pixel values within this box to 0) and then measure how much the prediction accuracy drops. This approach allows us to identify key regions affecting the classifier output. For example, if an image region is truly unimportant, then we should not see a change in the classifier output (e.g., the saliency map would have zero values for this region). Conversely, if an image region contains important syndromic features related to a certain genetic condition, then removing this region would decrease the prediction probability of this condition. In this case, the image region is considered to be positively contributing toward the prediction of the genetic condition (e.g., the saliency map would have nonzero values for this region).

We note that, our goal is to observe: (1) which regions of an image affect the prediction probability of the ground-truth label; (2) how these regions differ from those areas viewed most by human clinicians. Hence, we did not conduct the saliency analyses on the correct predictions and then on the incorrect predictions, separately. Rather, the saliency analyses were performed with respect to the ground-truth labels regardless of whether or not the model correctly identifies these ground-truths.

A saliency map for each test image with respect to its ground-truth label was taken as the average of the saliency maps from each of the models from 5-fold cross validation. As mentioned, for a specific test image, this ground-truth label may or may not be the model's correct prediction. We only require that the model's predicted probability of this ground-truth to not be very close to zero; otherwise, the saliency map may not be reliable when the model fails to make any prediction for the ground-truth [15]. When comparing to a human visual heat map, we applied a coverage-threshold to the saliency map. Pixel values below this coverage-threshold are set to zero; grid-search was used to find the best coverage-threshold where the remaining number of non-zero pixels is roughly equal to that of the human visual heat map. This coverage-threshold ensures that the human visual heat map and model saliency map are contributing equal numbers of "active" pixels toward the computation of the IoU and KL metrics; otherwise, these metrics can be unreliable.

We observed that, for human visual attentions at the low noise-threshold, the number of nonzero pixels takes up about 6–8% of the face region for all the test images. Hence, for each saliency map, the coverage-threshold was set to have 7% of the face region to be nonzero pixels. At a high noise-threshold for the human visual attentions, 3–4% of the face region were nonzero pixels, and thus the coverage-threshold for each saliency map was set to have 3.5% of the face region to be nonzero pixels.

We acknowledge that there are other models besides EfficientNet-B4 that have been studied in the context of genetic and other congenital diseases [4,5]. Moreover, there are other saliency approaches besides the occlusion method. This paper's combination of EfficientNet-B4 and the occlusion method seems to return reasonable results. In future studies, we plan to evaluate other types of image classifiers and saliency approaches.

Supporting information

S1 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original 22q11DS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S2 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original BWS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S3 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original CdLS image (see **S1 Text** for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S4 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original DS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S5 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original KS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S6 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original NS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S7 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original PWS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S8 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original RSTS1 image (see **S1 Text** for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S9 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original WHS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S10 Fig. (a) Eye gaze heat map of the participants overlayed on a segmented version of the original WS image (see <u>S1 Text</u> for original image sources). The leftmost image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a high threshold that more clearly indicates facial regions with high visual attentions (right). (b) Saliency maps of image with key regions that affect the classifier accuracy. We apply a low (left) and high coverage-threshold (right) on the saliency maps so that the area covered is approximately the same as the low and high noise-threshold, respectively. (TIF)

S11 Fig. IoU and KL metric comparing the visual attention of successful clinicians and model saliency maps over all the 10 test diseases (which includes RSTS1). A low and high noise-threshold were applied to the clinician heat maps, removing spurious and retaining only high-est visual activity, respectively. At each noise-threshold of a particular human visual map, an independent coverage-threshold was applied to the corresponding model saliency map. Pixels of a saliency map below this coverage-threshold were set to zero, and the remaining number of nonzero pixels would be roughly equal to that of the corresponding human visual map. Coverage-threshold ensures that a saliency map has equal number of positive pixels as its

corresponding human visual map. (TIF)

S12 Fig. IoU and KL metric comparing the heat maps (at different noise-threshold) of successful clinicians versus non-clinicians conditioned on whether the test images are of affected or unaffected individuals. On average, there is no major differences between how human participants view images of affected and unaffected individuals. (TIF)

S13 Fig. Eye gaze heat map of the participants overlayed on the unaffected images. For each set, the left most image is the average heat map of the successful clinicians (top) and non-clinicians (bottom). We apply two different noise-thresholds: a low threshold to remove possibly spurious visual interests (middle), and then a higher threshold that more clearly indicates facial regions with high visual attentions (right). (TIF)

S14 Fig. We defined the AOIs specific to the HPO-annotated features for the 22q11DS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see <u>S1 Text</u> for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S15 Fig. We defined the AOIs specific to the HPO-annotated features for the BWS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see <u>S1 Text</u> for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation.

(TIF)

S16 Fig. We defined the AOIs specific to the HPO-annotated features for the CdLS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see **S1 Text** for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S17 Fig. We defined the AOIs specific to the HPO-annotated features for the DS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see **S1 Text** for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S18 Fig. We defined the AOIs specific to the HPO-annotated features for the KS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see **S1 Text** for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S19 Fig. We defined the AOIs specific to the HPO-annotated features for the NS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see <u>S1 Text</u> for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S20 Fig. We defined the AOIs specific to the HPO-annotated features for the PWS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see **S1 Text** for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S21 Fig. We defined the AOIs specific to the HPO-annotated features for the RSTS1 image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see <u>S1 Text</u> for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S22 Fig. We defined the AOIs specific to the HPO-annotated features for the WHS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see S1 Text for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S23 Fig. We defined the AOIs specific to the HPO-annotated features for the WS image (left image). A segmented version of the original image is shown, and the AOIs drawn on the original image may not perfectly match the segmented version shown (see **S1 Text** for original image sources). Boxplots compare duration-of-fixation and time-to-first-whole-fixation. (TIF)

S24 Fig. For each of the 22 clinicians, we average heat maps over all 10 affected (a) and 6 unaffected (b) images. Each participant displays unique behavior. For example, in (a) the first participant (row 1 column 1) and sixth participant (row 2 column 1) showed unequal interest at the nose area.

(TIF)

S25 Fig. For each of the 22 non-clinicians, we average heat maps over all 10 affected (a) and 6 unaffected (b) images. Each participant displays unique behavior; however, visually this uniqueness is less detectable compared to **S24 Fig.** (TIF)

S26 Fig. We applied a low noise-threshold to remove spurious signals from the visual heat maps, and then estimated the differences in the visual attention between affected and unaffected images conditioned on the same participant (see <u>S24</u> and <u>S25</u> Figs). Participants were given random IDs of gene names. IoU and KL metric show that a few participants (e.g., clinician EP300) are more consistent with their visual behavior when viewing affected and unaffected images, suggesting that these participants may be looking at similar facial features for both types of images.

(TIF)

S27 Fig. We take the average of all the visual attention heat maps across all test images for the clinician (left) and non-clinician groups (right). We observed that most of the visual interests align with eyes, nose, and mouth areas. To account for normal human behavior when viewing an image, conditioned on the group expertise (clinician or non-clinician), we subtracted these common average areas from each individual heat map used for our analyses. This helps account for typical human behavior when viewing faces but does not cause us to ignore these areas of common facial attention.

(TIF)

S1 Text. Contains **Tables A-C** and additional explanations about the DL model analyses and image sources.

(DOCX)

Acknowledgments

This work utilized the computational resources of the NIH HPC Biowulf cluster.

Author Contributions

- **Conceptualization:** Dat Duong, Shahida Moosa, Susan Persky, Cedrik Tekendo-Ngongang, Tzung-Chien Hsieh, Rebekah L. Waikel, Elisabeth André, Peter Krawitz, Benjamin D. Solomon.
- **Data curation:** Dat Duong, Anna Rose Johny, Suzanna Ledgister Hanchard, Kendall Flaharty, Fabio Hellmann, Ping Hu, Behnam Javanmardi, Tanviben Patel, Cedrik Tekendo-Ngongang, Hellen Lesmann, Rebekah L. Waikel, Benjamin D. Solomon.
- Formal analysis: Dat Duong, Kendall Flaharty, Rebekah L. Waikel.
- Funding acquisition: Benjamin D. Solomon.
- **Investigation:** Anna Rose Johny, Suzanna Ledgister Hanchard, Christopher Fortney, Kendall Flaharty, Fabio Hellmann, Ping Hu, Hellen Lesmann, Tzung-Chien Hsieh, Rebekah L. Waikel, Benjamin D. Solomon.
- Methodology: Dat Duong, Anna Rose Johny, Suzanna Ledgister Hanchard, Christopher Fortney, Behnam Javanmardi, Shahida Moosa, Ömer Sümer, Cedrik Tekendo-Ngongang, Tzung-Chien Hsieh, Benjamin D. Solomon.
- **Project administration:** Anna Rose Johny, Suzanna Ledgister Hanchard, Christopher Fortney, Rebekah L. Waikel, Peter Krawitz.
- **Resources:** Suzanna Ledgister Hanchard, Christopher Fortney, Peter Krawitz, Benjamin D. Solomon.
- Software: Dat Duong, Kendall Flaharty, Ömer Sümer.
- **Supervision:** Dat Duong, Suzanna Ledgister Hanchard, Christopher Fortney, Susan Persky, Tzung-Chien Hsieh, Rebekah L. Waikel, Elisabeth André, Peter Krawitz, Benjamin D. Solomon.
- Validation: Dat Duong, Anna Rose Johny, Kendall Flaharty.
- Visualization: Dat Duong, Anna Rose Johny, Suzanna Ledgister Hanchard, Christopher Fortney, Kendall Flaharty, Benjamin D. Solomon.
- Writing original draft: Dat Duong, Benjamin D. Solomon.
- Writing review & editing: Dat Duong, Anna Rose Johny, Suzanna Ledgister Hanchard, Christopher Fortney, Kendall Flaharty, Fabio Hellmann, Ping Hu, Behnam Javanmardi, Shahida Moosa, Tanviben Patel, Susan Persky, Ömer Sümer, Cedrik Tekendo-Ngongang, Hellen Lesmann, Tzung-Chien Hsieh, Rebekah L. Waikel, Elisabeth André, Peter Krawitz, Benjamin D. Solomon.

References

- Lee P, Bubeck S, Petro J. Benefits, Limits, and Risks of GPT-4 as an AI Chatbot for Medicine. N Engl J Med. 2023; 388(13):1233–9. https://doi.org/10.1056/NEJMsr2214184 PMID: 36988602
- Jumper J, Evans R, Pritzel A, Green T, Figurnov M, Ronneberger O, et al. Highly accurate protein structure prediction with AlphaFold. Nature. 2021; 596(7873):583–9. https://doi.org/10.1038/s41586-021-03819-2 PMID: 34265844
- Cheng J, Novati G, Pan J, Bycroft C, Zemgulyte A, Applebaum T, et al. Accurate proteome-wide missense variant effect prediction with AlphaMissense. Science. 2023; 381(6664):eadg7492. https://doi.org/10.1126/science.adg7492 PMID: 37733863
- Hsieh TC, Bar-Haim A, Moosa S, Ehmke N, Gripp KW, Pantel JT, et al. GestaltMatcher facilitates rare disease matching using facial phenotype descriptors. Nat Genet. 2022; 54(3):349–57. https://doi.org/ 10.1038/s41588-021-01010-x PMID: 35145301
- Porras AR, Rosenbaum K, Tor-Diez C, Summar M, Linguraru MG. Development and evaluation of a machine learning-based point-of-care screening tool for genetic syndromes in children: a multinational retrospective study. Lancet Digit Health. 2021. https://doi.org/10.1016/S2589-7500(21)00137-0 PMID: 34481768
- Krupinski EA. Visual scanning patterns of radiologists searching mammograms. Acad Radiol. 1996; 3 (2):137–44. https://doi.org/10.1016/s1076-6332(05)80381-2 PMID: 8796654
- Bond RR, Zhu T, Finlay DD, Drew B, Kligfield PD, Guldenring D, et al. Assessing computerized eye tracking technology for gaining insight into expert interpretation of the 12-lead electrocardiogram: an objective quantitative approach. J Electrocardiol. 2014; 47(6):895–906. <u>https://doi.org/10.1016/j.jelectrocard.2014.07.011</u> PMID: 25110276
- Sudin E, Searjeant M, Partridge G, Phillips P, Hiller L, Snead D, et al. Digital pathology: the effect of experience on visual search behavior. J Med Imaging (Bellingham). 2022; 9(3):035501. <u>https://doi.org/ 10.1117/1.JMI.9.3.035501 PMID: 35572382</u>
- 9. Duong D, Solomon BD. Analysis of large-language model versus human performance for genetics questions. Eur J Hum Genet. 2023.
- DeGrave AJ, Janizek JD, Lee S-I. Al for radiographic COVID-19 detection selects shortcuts over signal. Nature Machine Intelligence. 2021; 3(7):610–9.
- Ledgister Hanchard SE, Dwyer MC, Liu S, Hu P, Tekendo-Ngongang C, Waikel RL, et al. Scoping review and classification of deep learning in medical genetics. Genet Med. 2022; 24(8):1593–603. https://doi.org/10.1016/j.gim.2022.04.025 PMID: 35612590
- 12. Deng J, Guo J, Zhou Y, Yu J, Kotsia I, Zafeiriou S. Retinaface: Single-stage dense face localisation in the wild. arXiv preprint arXiv:190500641. 2019.
- Ehteshami Bejnordi B, Veta M, Johannes van Diest P, van Ginneken B, Karssemeijer N, Litjens G, et al. Diagnostic Assessment of Deep Learning Algorithms for Detection of Lymph Node Metastases in Women With Breast Cancer. JAMA. 2017; 318(22):2199–210. https://doi.org/10.1001/jama.2017. 14585 PMID: 29234806
- 14. Wu JT, Wong KCL, Gur Y, Ansari N, Karargyris A, Sharma A, et al. Comparison of Chest Radiograph Interpretations by Artificial Intelligence Algorithm vs Radiology Residents. JAMA Netw Open. 2020; 3 (10):e2022779. https://doi.org/10.1001/jamanetworkopen.2020.22779 PMID: 33034642
- Saporta A, Gui X, Agrawal A, Pareek A, Truong SQ, Nguyen CD, et al. Benchmarking saliency methods for chest X-ray interpretation. 2022; 4(10):867–78.
- Gehrung M, Crispin-Ortuzar M, Berman AG, O'Donovan M, Fitzgerald RC, Markowetz F. Triage-driven diagnosis of Barrett's esophagus for early detection of esophageal adenocarcinoma using deep learning. Nat Med. 2021; 27(5):833–41. https://doi.org/10.1038/s41591-021-01287-9 PMID: 33859411
- Duong D, Waikel RL, Hu P, Tekendo-Ngongang C, Solomon BD. Neural network classifiers for images of genetic conditions with cutaneous manifestations. HGG Adv. 2022; 3(1):100053. https://doi.org/10. 1016/j.xhgg.2021.100053 PMID: 35047844
- Duong D, Waikel RL, Hu P, Tekendo-Ngongang C, Solomon BD. Neural network classifiers for images of genetic conditions with cutaneous manifestations. Human Genetics and Genomics Advances. 2021:100053. https://doi.org/10.1016/j.xhgg.2021.100053 PMID: 35047844
- Duong D, Hu P, Tekendo-Ngongang C, Hanchard SEL, Liu S, Solomon BD, Waikel RL. Neural Networks for Classification and Image Generation of Aging in Genetic Syndromes. Front Genet. 2022; 13:864092. https://doi.org/10.3389/fgene.2022.864092 PMID: 35480315
- Nguengang Wakap S, Lambert DM, Olry A, Rodwell C, Gueydan C, Lanneau V, et al. Estimating cumulative point prevalence of rare diseases: analysis of the Orphanet database. Eur J Hum Genet. 2020; 28 (2):165–73. https://doi.org/10.1038/s41431-019-0508-0 PMID: 31527858

- Brams S, Ziv G, Levin O, Spitz J, Wagemans J, Williams AM, Helsen WF. The relationship between gaze behavior, expertise, and performance: A systematic review. Psychol Bull. 2019; 145(10):980– 1027. https://doi.org/10.1037/bul0000207 PMID: 31414844
- 22. Tatler BW, Baddeley RJ, Gilchrist ID. Visual correlates of fixation selection: effects of scale and time. Vision Res. 2005; 45(5):643–59. https://doi.org/10.1016/j.visres.2004.09.017 PMID: 15621181
- Tan M, Le Q, editors. Efficientnet: Rethinking model scaling for convolutional neural networks. International Conference on Machine Learning; 2019 2019: PMLR; 2019.
- 24. Kingma DP, Ba J. Adam: A method for stochastic optimization. arXiv preprint arXiv:14126980. 2014.
- 25. Zeiler MD, Fergus R, editors. Visualizing and understanding convolutional networks. Computer Vision– ECCV 2014: 13th European Conference, Zurich, Switzerland, September 6–12, 2014, Proceedings, Part I 13; 2014: Springer